Notes for the RICH1 phone meeting 11th May 2007

People present:

Bristol:  Richard.

CERN: Kenneth, Nick, Fabio (part), Olav (part), Tim (part).

Imperial:  Bill, Dave, Gabriel, and Richard

Oxford: Neville, Rohan.
1. Beam Pipe Commissioning:

The commissioning of the beam pipe is delayed by three weeks, as a delivery is needed for the gas transfer line.  As a result, the installation of the mirrors and the temperature sensor monitors cannot begin until 24th June.  Another result is that there is now access to RICH1 in the coming week, although it is not clear if this is useful.

2.  Spherical Mirrors: 

2.1. Mechanics:

The feet for the spherical mirrors’ C-frame have been fitted to the frame.  Bill has the stretchable carbon fibre rods for the C-frames; they will be brought out and fitted next week.   The last remaining items are the stops that are needed for the mounting process.  These will not be needed until the 25th June and will be ready by then.

2.2 Installation trials:

The C side frame has been put in without problem.  The clearances between outside of the frame and the edge of the door bosses have been measured and found to be somewhat tighter than expected: <1mm rather than ~3mm.  This is due to CMA having glued a carbon fibre logo at the crucial part of the frame.  This may mean that the bosses need to be machined, which is probably preferable to grinding off the logo.  This will not be done in situ.  The machining may result in a leak via the door bolt holes, although this would not be a great problem to rectify.

2.3 Optical alignment.

Preparation of the optical bench support, which, along with the alignment rig are now in position in building 156 and are not to be moved.  The tripods which will be used for the surveying are also in place.  These will also be used for the flat mirror surveying.  The alignment and survey will begin around the end of May.

3. Flat mirrors:

3.1  Installation trials:
The lower flat mirror plates have been put into the gas enclosure.  The frame and box clash with the gas enclosure.  Also, the bridge rails were too short and the angle adjuster was mounted on the wrong side of the first-inserted plate.  All these problems were overcome and the two plates were mounted and mated together and left supported on the three support points, where they seemed very rigid.  The mating of the two plates proved to be the most difficult part of the task, although useful things have been learned from doing this.  Mechanical changes have also been made by Tim as a result of the trials.  These have been completed and a successful installation was repeated on the alignment rig.
3.2 Optical alignment:

Last Wednesday the survey team surveyed the back plates.  Next Tuesday the two lower plates will be put on and then surveyed.  On Monday cleaning of building 156 will take place, after which shoe covers will need to be worn.  The last remaining mechanical items needed for the alignment are the V-blocks, which should be in CERN by Sunday.  These will allow the upper flat mirror plates to be put in place.  Tim is coming to CERN again for the 21st to the 25th May, during which time the mirrors will be mounted on the plates (already inserted into their boxes).  A process of repeated mounting and dismounting will be undertaken to test stability.

4.  HPD assembly:

4.1 Column dressing:

No update from before:  Restart work on first half next week. Attach cooling plates and test temperature sensors on columns in next two weeks.  Column mounting from 4th June.

4.2 Upper rail:

Manufacturing is being done at NAB with delivery due at the end of May.  A trial fit is to be done in Oxford after that.  All other orders for rail pieces have been made.  Things are ready for a safety review next Wednesday, with all paper work in place.  Rohan and Bill will attend. A spreader for the crane-handling of the assembly might be needed.

4.3 Lower rail system:

This is more complicated as the lower HPD system has its electronics pointing down leaving little clearance and as a result rotation is needed in order to remove the columns.  A concept for the system has been agreed on for the system, with all the constraints having been discussed with Brian who is doing the design.  The installation rig at pit 8 is to be used also as an assembly rig, with the ladders to be inserted into the rig with the boxes vertical.   A further constraint comes from the angle of the HPD box which means that when it is rotated to vertical there is little space to take out the ladders because of the TT rail system.  A second rotation can solve this.  An update of Brian’s concept is awaited, although he may require more information on the HPD box design than there currently is.

4.4 Mechanics:

The frame is currently being machined.  Two end and two side pieces to be repeated four times and a hood for light tightness.

5. Cabling:

Scaffolding is being put up as of today for the cabling.  The upper cabling will be put in  the next two weeks by Ken and Johan, assisted by Richard and Chris Eames. Richard will supply cardboard template of appropriate patch panels.
6. Light leak detector:

Photo multipliers of the required specifications (available woth bialkali or multialkali cathodes) have been found, although further information is needed from the manufacturers about exactly what comes with the devices, i.e. whether anything else would needed to be made or purchased before they could be used.

7. Photon funnel:

Quotation has been received from the company involved with the RICH2 photon funnel.

8.Magnetic distortion monitoring system:

Ray has been given all the information he needs to proceed, that is, information to allow specifying the LED focussing and on allowed mechanical dimensions.  Syracuse have found a way to purchase the stage.  The next meeting with them is on Tuesday the 22nd of May.

9.  LAMS:

Testing of the three cameras using a single controller at CERN next week by Richard.

10.  Aerogel mounting trials:

Took place but were restricted by beam pipe.  If further tests are needed these are best done after flat mirror installation.

11.  Other business:

The magnet will not be powered before the end of August.

The RICH1 box manufacturers have halved their requested extra payment.  A decision on whether to accept this is required soon.

The temperature sensors should be installed before the flat mirror installation.
